Author's response to reviews

Title: Prosthetic valve endocarditis caused by Pseudomonas luteola

Authors:

Jean-Paul Casalta (jean-paul.casalta@wanadoo.fr)
Pierre-Edouard Fournier (Pierre-Edouard.Fournier@medecine.univ-mrs.fr)
Gilbert Habib (gilbert.habib@ap-hm.fr)
Alberto Riberi (alberto.riberi@ap-hm.fr)
Didier Raoult (didier.raoult@medecine.univ-mrs.fr)

Version: 3 Date: 13 September 2005

Author’s response to reviews:

Dear Mr Holmes,

Attached, please find the revised version of our manuscript: Prosthetic valve endocarditis caused by Pseudomonas luteola. The manuscript was modified according to your recommendations.

1-In particular, we made sure that the Title & Abstract were identical in the manuscript and in the submission system.

2-Author list - The list of authors in the manuscript was written exactly as they are in the submission system (forename, initial (if any), surname, in sentence case).

3-Title - We used sentence case in the title, both in the manuscript file and the submission system.

4-The word count was removed from the manuscript.

5-"Informed consent was obtained from the patient" was moved to the acknowledgement section.

6-Section headings - We used sentence case for all headings and sub-headings in the manuscript.

7-Abstract - The abstract was structured into Background, Case Presentation, and Conclusions. We also updated the abstract details on the submission page.

8-Highlighting/Tracking was removed from the manuscript text.

9-Manuscript sections - Manuscript sections now include (in the following order): Abstract; Background; Case Presentation; Conclusions; Competing interests; Authors' contributions; Acknowledgements; References.

Please, do not hesitate to contact me for any further comment or question.

Yours, sincerely

Jean-Paul Casalta